Author's response to reviews

Title: "I should live and finish it": A qualitative inquiry into Turkish women's menopause experience

Authors:

Serap Y Cifcili (serapcifcili@gmail.com)
Mehmet Akman (makman4@gmail.com)
Abdullah Demirkol (demirkolster@gmail.com)
Pemra Unalan (pcunalan@gmail.com)
Etienne Vermeire (etienne.vermeire@ua.ac.be)

Version: 2 Date: 31 August 2008

Author's response to reviews:

I corrected a few punctuation and spelling mistakes.
I added the following sentence to the method in page 6, third paragraph; It is important to note that, in Turkey, primary care does not have a “gate-keeper role”.
I deleted the sentence referring participants' age from the method part since this information was also given in the results.
I changed the first sentence in page 15 (results) as follows;
"On further questioning, some of the participants who did not seek help were also being medically checked in terms of routine screening."
I deleted the following sentence from the limitation part, because this information is being constantly changing because of the new regulations in health insurance system.
"Approximately 35% of Turkish population has no health insurance coverage and are not able to attend such clinics".